

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Grimm, Michael; Hartwig, Renate; Reitmann, Ann-Kristin; Bocoum, Fadima Yaya

# Working Paper Can informal redistribution withstand formal safety nets? Insights from urban-rural transfers in Burkina Faso

Passauer Diskussionspapiere - Volkswirtschaftliche Reihe, No. V-81-20

**Provided in Cooperation with:** University of Passau, Faculty of Business and Economics

*Suggested Citation:* Grimm, Michael; Hartwig, Renate; Reitmann, Ann-Kristin; Bocoum, Fadima Yaya (2020) : Can informal redistribution withstand formal safety nets? Insights from urban-rural transfers in Burkina Faso, Passauer Diskussionspapiere - Volkswirtschaftliche Reihe, No. V-81-20, Universität Passau, Wirtschaftswissenschaftliche Fakultät, Passau

This Version is available at: https://hdl.handle.net/10419/224947

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU



# Can informal redistribution withstand formal safety nets? Insights from urban-rural transfers in Burkina Faso

Michael Grimm, Renate Hartwig, Ann-Kristin Reitmann, Fadima Yaya Bocoum

Diskussionsbeitrag Nr. V-81-20

Volkswirtschaftliche Reihe ISSN 1435-3520

# PASSAUER DISKUSSIONSPAPIERE

Herausgeber: Die Gruppe der volkswirtschaftlichen Professoren der Wirtschaftswissenschaftlichen Fakultät der Universität Passau 94030 Passau

# Can informal redistribution withstand formal safety nets? Insights from urbanrural transfers in Burkina Faso

# Michael Grimm, Renate Hartwig, Ann-Kristin Reitmann, Fadima Yaya Bocoum

Diskussionsbeitrag Nr. V-81-20

# Volkswirtschaftliche Reihe ISSN 1435-3520

Adresse des Autors/der Autoren:

Ann-Kristin Reitmann Wirtschaftswissenschaftliche Fakultät Universität Passau 94030 Passau

Telefon: +49 851 509 5313 Telefax: +49 851 509 3312 E-Mail: ann-kristin.reitmann@uni-passau.de

Für den Inhalt der Passauer Diskussionspapiere ist der jeweilige Autor verantwortlich. Es wird gebeten, sich mit Anregungen und Kritik direkt an den Autor zu wenden.

# Can informal redistribution withstand formal safety nets? Insights from urban-rural transfers in Burkina Faso

Michael Grimm, University of Passau, IZA and RWI Research Network
 Renate Hartwig, GIGA Hamburg and University of Göttingen
 Ann-Kristin Reitmann\*, University of Passau
 Fadima Yaya Bocoum, Institute de Recherche en Sciences de la Santé (IRSS)

#### July 2020

Abstract: Households in rural areas still depend on informal transfers to meet subsistence needs and cope with shocks. Yet, to provide additional monetary support, formal safety nets are increasingly introduced in developing countries. However, it remains unclear whether such social protection policies will have the desired welfare effects. This article addresses this question by analyzing the private transfer response to changes in the income of rural recipients using novel data from Burkina Faso. We assume that the transfer-income relationship is a non-linear one where transfer motives, and therefore also transfer responses, vary with the recipient's position in the income distribution. Our findings support this view. We find a pronounced, negative private transfer response among the poorest of the poor. This observation has important policy implications, because those households that depend most on private transfers, would be most affected by crowding-out effects. In terms of transfer motives, the negative relationship for the lowest income class is consistent with transfers being altruistically motivated. With increasing income levels, transfers cease being altruistic at the margin and switch toward exchange motives. However, the observed transfer pattern is also indicative of an (informal) insurance role of private transfers. Rural households receive higher private transfers in response to negative shocks. These results can serve as a basis for the design of formal social protection mechanisms in a context where informal redistribution still plays an important role.

Key words: private transfers, crowding-out, sharing norms, informal insurance, Burkina Faso *JEL* codes: D64, H31, I30, O12

<sup>\*</sup>Corresponding author: Ann-Kristin Reitmann, Innstraße 29, 94032 Passau, Germany; annkristin.reitmann@uni-passau.de. This paper is part of the Formal Insurance and Productive Effects Study (FIdES) – a collaborative research project of ASMADE, the Institute de Recherche en Sciences de la Santé (IRSS) in Ouagadougou, the University of Passau and the DIE in Bonn. Funding from the Rotterdam Global Health Initiative, the Bavarian Research Alliance (03124520) and the DIE is gratefully acknowledged. Renate Hartwig further acknowledges financial support from the F.R.S-FNRS, Mandate No. 6335. The funding sources had no involvement in any stage of this study. We thank Christopher Udry and Harunan Kazianga for valuable comments on earlier versions of this paper and are grateful for comments received from seminar participants at the University of Kent as well as conference participants in Göttingen (GLAD 2018), Zurich (AEL 2018), Berlin (AEL Doctoral Workshop 2018), Munich (BGPE Doctoral Workshop 2019) and Oxford (CSAE 2019).

### 1 Introduction

Since the turn of the century, social safety net programs expanded rapidly around the world. To date, they cover about 2.5 billion people in low- and middle-income countries. Yet, many of the poor in these countries still remain excluded. In low-income countries, formal safety nets reach only 20% of the poor (World Bank, 2018). The remaining share has to rely on their social network for support. Hence, informal transfers still play a key role in helping people to cope with shocks and provide income support to those that are far below subsistence. These informal transfers come mostly from families and friends (Fafchamps & Gubert, 2007; Fafchamps & Lund, 2003; Lucas & Stark, 1985). The literature shows that the motives for informal transfers are manifold and strongly intertwined. Theory differentiates between altruism (see Becker, 1974) and exchange (see Bernheim, Shleifer, & Summers, 1985; Cox, 1987). In practice, however, these motives might be difficult to disentangle. Especially, because we generally are not looking at one transfer in particular, but at repeated interactions over the course of an observation period. Irrespective of the transfer motive, the fundamental and more general question is if and how (overall) transfers respond to the recipient's income. This reaction, in the literature called the "transfer derivative", is an important factor for policy decisions. It provides information about how formal social safety nets could crowd out private transfers and in the extreme case offset their intended redistributive effects completely. This would leave the safety net without effect (Cox, Hansen, & Jimenez, 2004).

In this study, we investigate informal transfers in Burkina Faso. Burkina Faso provides an interesting case study, because it exemplifies the context described above. It is one of the poorest countries in the world, holding position 182 out of 189 in UNDP's Human Development Index ranking (Grimm, Wetta, & Nikiema, 2016; UNDP, 2019). Formal social safety nets in form of cash transfers or large-scale insurance schemes are still absent. Yet, a number of programs are currently piloted. The World Bank, for example, introduced a cash transfer program (social safety net project) and provides support to the pilot of a national health insurance scheme. Hence, the question whether there might be crowding-out between public and private transfers is of relevance in this context, as it is in many other countries that are in comparable economic conditions.

We contribute to a large body of research on the issue of crowding-out effects. The seminal work on the transfer-income relation and the possibility of crowding-out effects goes back to Becker (1974) and Barro (1974). The theoretical model by Becker (1974), for instance, predicts that an increase in household resources is met with a decrease in altruistically motivated transfers. Transfers given with the purpose to mitigate risk would follow the same qualitative pattern (Cox et al., 2004). The model developed by Cox (1987) introduces the motive of exchange to the discourse, where transfers can be both positively or negatively related to household resources. While empirical evidence suggests that crowding-out is less of an issue in developed

countries (see e.g., Altonji, Hayashi, & Kotlikoff, 1997; Cox & Jakubson, 1995), crowding-out effects are observed in studies from developing countries. However, there is no uniform conclusion, particularly when looking at these effects along the income distribution. The transfer-income relationship exhibits non-linearities, where different transfer motives co-exist. Thus, crowding-out effects depend on the recipient's position in the income distribution. In the Philippines, Cox et al. (2004) find strong negative transfer derivatives only among low-income households. A similar pattern prevails in the study by Jimenez and Brown (2012) from Fiji. By contrast, Kazianga (2006) finds no evidence of crowding-out for the lower tail of the income distribution in Burkina Faso, but rather for the middle-income group.

More recent literature moved away from studying the transfer derivative and turned to focusing on cases where formal transfers (e.g., insurance) were actually introduced, or at least offered, and investigated the reaction of private transfers to this institutional change (e.g., Attanasio & Ríos-Rull, 2000; Geng, Janssens, Kramer, & van der List, 2018; Jensen, 2003; Mobarak & Rosenzweig, 2013; Strupat & Klohn, 2018). Others have studied this issue using lab-in-the field experiments, where potential confounders can be better controlled for, but where the external validity, that is, the transferability of the findings to situations outside the lab, is often subject to debate (e.g., Cecchi, Duchoslav, & Bulte, 2016; Landmann, Vollan, & Frölich, 2012; Lenel & Steiner, 2017; Lin, Liu, & Meng, 2014). The evidence from these studies is mixed. A large body of research though confirms crowding-out effects for social security, pensions, food aid and conditional cash transfers (Attanasio & Ríos-Rull, 2000; Cox & Jimenez, 1992; Dercon & Krishnan, 2003; Jensen, 2003). Moreover, crowding-out of informal transfers is also observed for insurance instruments such as the national health insurance scheme in Ghana (Strupat & Klohn, 2018) and a weather insurance in rural India (Mobarak & Rosenzweig, 2013). Yet, some studies do not find crowding-out effects. In rural Kenya, for example, Geng et al. (2018) conclude that formal and informal insurances are complements. Dercon, Hill, Clarke, Outes-Leon, and Seyoum Taffesse (2014) also reject crowding-out. In Ethiopia, they argue that the provision of index insurance rather crowds in informal sharing arrangements. The rejection of crowding-out for insurances often comes with the argument that formal insurance increases the resources in the risk-pool of informal sharing networks, and therefore ultimately renders these systems more efficient. This applies mainly to responses to covariate shocks such as droughts or flooding, which typically affect entire communities (Berg, Quirion, & Sultan, 2009; Dercon et al., 2014; Mobarak & Rosenzweig, 2013). This argument also holds for the more qualitative sociological literature (e.g., Heemskerk, Norton, & de Dehn, 2004). Some studies suggest that formal insurance increases the cohesion (e.g., Edin & Lein, 1997; O'Connor, Orloff, & Shaver, 1999), whereas others tend to argue that it is rather detrimental to it (e.g., Dercon & Krishnan, 2003; Ligon, Thomas, & Worrall, 2002).

We contribute to the existing literature by taking a step back again, using an ex-ante perspective on planned social safety net interventions. We believe this is important and has been overlooked in recent studies. This approach will provide policy-makers searching for guidance on effective policies with valuable and locally applicable insights and therefore offers a worthwhile complement to the existing ex-post evidence. Our data collected in the Central Plateau region of Burkina Faso is particularly suited to study this issue. First, our data set contains information about both ends of the transfer relationship. That is, for a sub-sample, we have information on both recipient and sender incomes. Especially the latter is often missing and may lead to a serious omitted variable bias. Second, our data follows the same households over time, which allows controlling for many other potential confounders. We use this data to model and estimate the transfer-income relationship similar to some of our predecessors in a non-linear way with spline specifications (see e.g., Cox et al., 2004; Kazianga, 2006). This allows us to explore not just whether crowding-out exists, but also how the income elasticity of transfers – or the transfer derivative – varies across the income distribution. This is particularly important for the targeting of social safety nets and the design of the transfer size relative to the recipient's income.

Our empirical results suggest that the relationship between the recipient's income and the transfer response is indeed non-linear. We find behavioral patterns that are theoretically consistent with altruism, effective risk-sharing, or both. Specifically, we observe a negative relation between the recipient's income and private transfers received for low-income households. This observation has important implications for the redistributive effectiveness of public transfers. It implies that attempts by the government or non-governmental organizations to redistribute income through formal interventions may potentially crowd out private responses. Especially poor households, which are also the ones that depend most on private transfers, would be most affected by crowding-out effects. This finding is in contrast to Kazianga (2006), who finds no evidence of crowding-out for the lower tail of the income distribution but rather for the middleincome group. Our results are robust to an instrumentation of the recipient's income. Moreover, our main findings also hold if we control for sender income in the transfer function to redress a potentially important omitted variable bias. Various papers raised the concern of a potential omitted variable bias in the transfer function due to missing information on sender income (Altonji et al., 1997; Cox & Jimenez, 1998; Cox & Rank, 1992; Kazianga, 2006). We can confirm that this bias is statistically relevant, yet, in our case the bias is quantitatively small.

Concerning the transfer patterns, we observe particularly high transfers in response to negative shocks, which also highlights the insurance role of informal transfers. When separating income into permanent and transitory components, we observe that (private) transfers only respond to transitory income. Assuming that our measure for transitory income is reliable, this would suggest that while formal insurance may imply crowding-out effects, other formal, more permanent forms of public transfers (such as cash transfers) might trigger another response.

The remainder of this paper is organized as follows. Section 2 presents the conceptual framework outlining the non-linear transfer response associated with different underlying motives. Section 3 describes the survey design and provides some details on the magnitude of transfers in our context. Section 4 outlines our empirical approach. The results as well as the robustness checks are discussed in Section 5. Section 6 concludes.

### 2 Conceptual framework

We largely follow Cox et al. (2004) in providing a conceptual framework and making the following two main points: First, we imply a non-constant transfer derivative with the possibility of two theoretical motives – altruism and exchange – to co-exist. Second, we acknowledge that the predictions for altruistically motivated transfers are analytically very similar to transfers for risk-sharing. Risk-sharing models, like the altruism-exchange framework can also imply nonlinear transfer derivatives.

#### 2.1 Altruism and exchange

In order to relate transfers received to recipient income, we apply a variant of the model by Cox (1987). This framework can also be used to display a non-linear pattern of received transfers in recipient income, where the pattern is determined by the motive prevailing at the margin. We believe altruism to operate at the margin when the recipient experiences financial hardship; that is, at very low levels of income with the donor not expecting any return. Empirically, this translates into a negative relationship between transfers received and recipient income. Thus, transfers are falling with increasing household resources. This prediction gives rise to the possibility that private transfers are crowded out by public ones, as the latter usually aim to increase household incomes experiencing financial hardship. According to the standard Becker-Barro model (1974), transfers would stop upon reaching the limits of altruism. In turn, the framework by Cox (1987) augments the altruistic model to include the possibility of interhousehold exchange. It allows for transfers to cease being altruistic at the margin and to switch to exchange related transfers. This motive switch may occur after recipient income has risen to a certain, though mostly unknown threshold. In contrast to altruistically motivated transfers, transfers motivated by exchange do not need to have a strong (negative) transfer derivative – in fact, the transfer derivative might even be positive. Formally, we assume the following utility of the donor d:

$$U_d = U(C_d, s, V(C_r, s)), \tag{1}$$

where V represents the recipient's r utility function. Both the donor's and the recipient's utility function depends on the consumption C of an aggregate good and the amount of services s provided or consumed. The latter may stand for anything that the recipient provides to the donor such as help with home production or future financial transfers. It is assumed that these services have no market substitutes. The budget constraints for the donor and the recipient are  $C_d = I_d - T$  and  $C_r = I_r + T$ , respectively, where I denotes pre-transfer incomes and T financial transfers given from the donor to the recipient. The utility expressed in Equation 1 captures both motives. On the one hand, the donor cares about the recipient so that  $\partial U_d/\partial V > 0$ . On the other hand, the recipient must be compensated for any services provided, because  $\partial V/\partial s < 0$ . All other partial effects of the remaining arguments in Equation 1 are positive. In order to see when a transfer is altruistically versus exchange motivated, one has to look at the participation constraint in Equation 2, which states that the recipient entering in the transfer relationship must not lower utility:

$$V(I_r + T, s) \ge V_0(I_r, 0)$$
 (2)

Cox (1987) shows that his model generates the two regimes of altruism and exchange, depending on whether the participation constraint is binding or not. First, if the constraint is not binding, transfers are altruistically motivated, and  $\partial T/\partial I_r < 0$  (i.e., transfers decrease with recipient income). Second, if the constraint is binding, transfers are exchange-related and exactly compensate for provided services. When motivated by exchange, the relationship between  $I_r$ and T is different. Cox (1987) shows that transfers first rise and then fall, thus, generating an inverted U-shaped relationship (see also Cox et al., 2004, for an illustration). The participation constraint, moreover, provides ground to explain why altruism prevails when recipient income is low, because in this case the constraint is not binding. Donors maintaining a transfer relationship with a poor household make transfers with the sole motivation to raise the household's well-being. Typically, it is assumed that the financial position of the donor outweighs the financial position of the recipient. Only when the recipient household achieves a certain level of well-being – or income (i.e., making the participation constraint binding) – exchange becomes the operative motive at the margin. Thus, the altruistic motivation might vanish, even though transfers themselves might not. Empirically, the described relationship needs to be addressed in a transfer function, which is non-linear in recipient income. This allows to (ex-ante) assess the possibility of crowding-out of private transfers by public ones. In this case, it also allows seeing which part of the population will be affected most. In our empirical specification, the non-linearity will take the form of (connected) splines.

#### 2.2 Risk-sharing and the insurance motive

Other commonly used models to investigate household transfers are risk-sharing models (Cochrane, 1991; Townsend, 1994). Also Becker (1974) notes that operative, altruistic transfers can imply effective risk-sharing between the transfer participants. However, altruism is not a necessary condition for such an informal insurance mechanism. Still, transfers related to altruism and risk-sharing are analytically very close, making their implications and empirical predictions very similar (see Cox et al., 2004). First, if transfers are used to cope with shocks that result in decreased household resources, then transfers and recipient income should be inversely related. Assuming that poorer households are more likely to experience negative income shocks and are more in need for (informal) insurance payments to cope, one would expect that transfers are

particularly responsive to income at low levels of recipient income under full risk-sharing. By contrast, households higher up in the income scale may also experience negative shocks, but are more likely to rely on savings, formal insurance or credit to cope. It follows that household risk-sharing models can also imply non-linear transfer responses. The possibility of crowding-out is just as relevant, considering that formal insurance or greater access to capital markets may interact with informal insurance payouts.

In our empirical analysis, following Kazianga (2006), we introduce permanent and transitory income in our transfer function. This specification allows us to explore the role of transfers in risk-sharing. If (full) risk-sharing is taking place, transfers should only be determined by the transitory part of income (Cochrane, 1991; Mace, 1991; Townsend, 1994).

Although transfer derivatives strongly depend on the transfer motive (which in turn depends on recipient income), our main goal is not to unambiguously identify the underlying transfer motives. We use the motives only to introduce predictions on the relationship between transfers received and recipient income. Finding that transfers respond negatively to the recipient's income is not enough to distinguish between altruism and risk-sharing. Moreover, the empirical identification of the exchange motive is particularly challenging when reciprocity is deferred in time, because then repeated observations of the same household over longer time periods are necessary and life-cycle considerations matter. This is, for example, the case for intergenerational exchange where parents support their children and the latter reciprocate when in adulthood to provide old-age support (Cox & Rank, 1992). An extreme case is "generalized reciprocity" (La Ferrara, 2003), where the offspring of the original benefactors is involved for reciprocity. An example would be young people receiving support from older relatives to finance their studies, whereas these young people in old age reciprocate by helping their younger relatives once they start earning money (La Ferrara, 2010).

# 3 Survey design and sample description

#### 3.1 Survey design

The empirical analysis is based on two related sets of household panel data: rural and urban. The rural household data was collected among households in the community of Ziniaré, which is located in the Plateau Central region of Burkina Faso. The rural community spans over 48 villages and is home to a bit less than 6,800 households, most of them engaged in subsistence farming. Our sample covers 2,007 randomly selected households from 38 villages. The data represents 32% (i.e., 14,164 individuals) of the population living in the area.

The first wave of data collection was conducted in October and November 2013, where we interviewed 1,499 households. With additional financial resources becoming available, we extended the first wave by another 508 households in December 2014. From the 2,007 interviewed

households, we ended up having full information from 1,996 households. The survey included modules on socio-economic household characteristics, food and non-food consumption, shocks, and transfers given and received in the 12 months prior to the survey, including information on the respective transfer recipient and sender. In March/April 2016, we were able to re-interview 1,876 households for the second wave.<sup>1</sup>





The urban household panel is linked to the rural household panel through the information about transfers received and given in the first wave of the rural household survey (2013/14). In that survey we collected detailed information on the amount, regularity and purpose of the transfer, and also personal characteristics of the sender (age, relationship, occupation, place of residence). Furthermore, we recorded if the sender was a business owner in an urban location in Burkina Faso. If this was the case, we asked for his or her contact details.<sup>2</sup> From the rural household survey, we collected information on 402 urban transfer sending households. We were able to trace and interview 359 of them in February/March 2014 and January 2015. The majority (85%) of them reside in the capital Ouagadougou. 95% of the senders are blood-related, being very often either a child or a sibling of the rural household head. In the second round of data collection in May/June 2016, we were able to re-interview 225 of the 359 households.<sup>3</sup> Figure 1

<sup>&</sup>lt;sup>1</sup>The follow-up was planned for October 2015, but had to be postponed due to the political instability in the country. The attrition rate is 6%. We test for systematic attrition and find that the attrited households differ slightly in their demographic composition, female headship and chronic disease prevalence (for details see Appendix A). We account for these differences in all our regressions.

<sup>&</sup>lt;sup>2</sup>The focus on business owners in the survey stems from the objective of the FIdES project, from which the data stems from. It had been developed with the objective of investigating the effects of transfer relationships on small and micro entrepreneurs.

<sup>&</sup>lt;sup>3</sup>Attrition in the urban sample is with 37% relatively high. A test for systematic attrition shows that households that drop out differ in their demographic composition and are less likely to have suffered from

summarizes the composition of our samples across waves.

For the linked sample, we form dyads for both waves between the urban households and their rural transfer recipient. While we work with balanced panels for the descriptive and empirical analysis when looking at the rural and urban sample separately, we use an unbalanced panel in our empirical analysis with the linked sample (to not lose any observations and gain statistical power).

#### 3.2 Sample description

We describe the rural and urban sample in Table 1. It becomes apparent that the rural households in our sample have on average a lower annual income and larger families than the urban households. The urban household heads have higher levels of education and around half own a business. However, when it comes to the households' vulnerability, both household samples seem comparable. Around 31% of the rural households have a chronically sick household member and with 26%, this share is only slightly smaller among urban households. In both samples, around 36% have experienced a negative shock in the 12 months prior to the survey conducted in the first wave. In the following, we take a closer look at the transfer behavior (i.e., transfers received and given) of both household samples.

We focus on transfers between households and ignore transfers within households. Transfers reported can be in cash or in-kind. In the following and in the empirical analysis we do not differentiate between both forms; in-kind transfers are assigned monetary values. Transfers were sent within a concentrated network, namely, largely among first- and second-degree relatives (parents, siblings, children). Only 9% of the transfers received and 12% of the transfers given came from or went to non-family members. This transfer pattern is in line with qualitative evidence from the anthropological literature. Hammond (1966) and Fiske (1990), for instance, stress the importance of "gift exchange" within the family as an integral part of the Mossi culture. We observe a similar concentration also in the urban sample. 92% of transfers given by urban households over the two survey waves went to first degree relatives, either from children to parents or vice versa. Yet, urban households reported a higher share (30%) of transfers received from donors outside of their family network.

Table 2 provides an overview of the transfer activity in our samples. More than half of the rural households do not take part in a transfer network. For those in transfer networks, we see that rural households are typically on the receiving side. By contrast, urban households are

a shock (for details see Appendix A). Our empirical analysis concentrates on the rural sample. Hence, the large attrition in the urban sample should not be a major concern. We test if rural households without a link in the second wave differ systematically from the other households. The results in Appendix A indicate no systematic differences with one exception: household heads in these households are slightly less likely to have secondary education. We control for this difference in our estimations with the linked sample.

	Rural HHs		Urbar	n HHs
	(1)	(2)	(3)	(4)
	Mean	SD	Mean	SD
Annual HH income (1,000 CFA F)	588.7	956.1	2,038.5	1,838.0
Livestock (TLU)	3.7	14.3		
Land owned (ha)	3.2	3.4		
HH size	7.2	3.6	5.6	3.1
No. children 0-5	1.3	1.2	1.0	1.0
No. children 6-18	2.7	2.1	1.8	1.7
No. adults 19-64	2.8	1.4	2.8	1.4
No. older adults 65+	0.3	0.6	0.1	0.3
Children/adult ratio	1.5	0.9	1.0	0.7
Age HH head (years)	48.5	14.8	41.7	12.0
HH head no schooling $(=1)$	0.720		0.405	
HH head primary school $(=1)$	0.118		0.222	
HH head secondary school $(=1)$	0.162		0.373	
Female HH head $(=1)$	0.060		0.040	
HH has business $(=1)$			0.524	
Chronically sick HH member $(=1)$	0.309		0.262	
Shock in past 12 month $(=1)$	0.365		0.360	
No. of HHs	1,876		225	

Table 1: Summary statistics, rural and urban households

*Note:* Data from first wave. Annual (net) income is approximated with annual household consumption net of transfers received and given. The value of livestock is expressed in tropical livestock units with the following conversion factor: cattle=0.7, sheep=0.1, goats=0.1, pigs=0.2 and chicken=0.01. A shock is defined as an event that caused, according to the household, serious problems for the living conditions of the household.

Table 2: Participation in informal transfers in the 12 months preceding the two survey waves, rural and urban households

	Rura	l HHs	Urban HHs		
	(1) Wave I (%)	(2) Wave II (%)	(3) Wave I (%)	(4) Wave II (%)	
Non-participants	53.2	52.3	11.6	15.1	
Recipients only	36.6	27.7	0.4	1.3	
Donors only	4.9	9.7	63.6	62.7	
Both recipient and donor	5.3	10.3	24.4	20.9	
No. of HHs	1,876	1,876	225	225	

more likely to be donors only.<sup>4</sup> Table 3 presents the transfer amounts (in 1,000 CFA F) and

<sup>&</sup>lt;sup>4</sup>The shares might be upward biased due to the sampling process. Also, note that some urban households reported not to have given transfers. This suggests a significant degree of misreporting from the urban or the rural side. Comola and Fafchamps (2017) and De Weerdt, Genicot, and Mesnard (2019)

variability over time. For this purpose, transfers were aggregated at the household level for cases where households reported more than one transfer given or received, respectively. In the first wave, 41.9% of the rural households received at least one transfer in the 12 months prior to the survey. The average transfer amounts to 72,100 CFA Franc (125.0 USD) or 12% of the average yearly household income. We see that transfer sizes are not stable over time; both transfers received and given are lower among the rural sample in the second wave. Moreover, the transfer amounts given and received by urban households are almost twice the amounts reported in the rural sample.

	Wave I		Wave II	
	(1) Transfer size (1,000 CFA F)	(2) No. of HHs	(3)Transfer size (1,000 CFA F)	(4) No. of HHs
Rural HHs				
Annual transfers received Annual transfers given Total no. of HHs	72.1 44.2	$785 \\ 190 \\ 1,876$	63.1 27.7	$712 \\ 375 \\ 1,876$
Urban HHs				
Annual transfers received Annual transfers given Total no. of HHs	46.3 113.6	$56 \\ 198 \\ 225$	34.1 137.9	50 188 225

Table 3: Size of informal transfers received and given in the 12 months preceding the two survey waves, rural and urban households

*Note:* Transfers received and given include all monetary and in-kind transfers that a household has received or given in the 12 months preceding the two survey waves; all in-kind transfers are assigned monetary values. Transfers received from various senders or given to various recipients are aggregated. The mean values in Columns 1 and 3 are conditional on receiving/giving a transfer. 1 USD corresponds to approx. 577 CFA F.

In our analysis, we focus on the rural household sample. First, because the simple description demonstrates that transfers are mainly unidirectional, that is, urban households are donors and rural households are recipients.<sup>5</sup> This may suggest that most rural households are not actively engaged in exchange. Second, in the case of Burkina Faso, rural households are the obvious target population for social safety nets. Hence, an ex-ante assessment is highly relevant in this context. Yet, taking a closer look at the rural households reveals that the recipients of the transfers described above are not solely concentrated in the poorest quintile. Within each income quintile, a share of 34-46% received a transfer (see Appendix B for details). The only distinction is that average transfers in the poorest income quintile are around double in size of those in

made similar observations.

<sup>&</sup>lt;sup>5</sup>A more detailed analysis of transfers received by rural households indeed confirms that a large proportion of transfers is coming from the urban area (see Appendix B). In the urban sample, a large share of reported transfers given are directed towards recipients living in the home village. In turn, urban households receive support largely from urban areas or abroad.

the other income quintiles. A fairly large amount of transfers received is in response to shocks, which points to the prevalence of risk-sharing. Around 50% of these negative shocks involve health shocks. Rural households also report to receive regular transfers; here, the prevailing reason is to meet subsistence needs (for further details see Appendix B).

## 4 Empirical strategy

In the empirical analysis, we estimate transfer response functions for the rural households, which relate transfers received to the income of the recipient household. As discussed in the conceptual framework, we assume that there is a non-linear relationship between transfer size and recipient income. We depict this non-linearity by using a regression spline model. It consists of continuous linear splines with four knots. The knots separate recipient income into income quintiles. Our approach differs from that of Cox et al. (2004), who only use one knot and treat this threshold level of income (where transfer behavior switches from altruistic to non-altruistic) as unknown. By contrast, Kazianga (2006) uses income quartile splines to depict the non-linearity, but he applies unconnected splines to do so. Our regression model takes the following form:

$$T_{ivt} = \sum_{k}^{5} \gamma_k I_{ivt} + \beta X_{ivt} + \alpha_t + \delta_v + u_{ivt}, \qquad (3)$$

where  $T_{ivt}$  is the annual amount of transfers received by rural household *i* in village *v* in the 12 months preceding the survey at time t.<sup>6</sup> *k* indicates the income quintile,  $I_{ivt}$  is household *i*'s annual (net) income at time *t* in linear spline *k*. The knots for creating the linear splines are specified to be placed at five percentiles of the income data. The interpretation of the coefficients  $\gamma_k$  is described in the following (income values in 1,000 CFA F – i.e., 1.73 USD):

$$\frac{\partial T_{ivt}}{\partial I_{ivt}} = \begin{cases} \gamma_1 \ if \ I_{ivt} < 214.1 \\ \gamma_2 \ if \ 214.1 \le I_{ivt} < 357.7 \\ \gamma_3 \ if \ 357.7 \le I_{ivt} < 540.1 \\ \gamma_4 \ if \ 540.1 \le I_{ivt} < 875.2 \\ \gamma_5 \ if \ I_{ivt} \ge 875.2 \end{cases}$$

Since we imply knots and do not treat them as unknown, the model can be estimated using (pooled) ordinary least squares (pOLS). Considering that some households do not participate in any transfer activity as shown in Table 2, we estimate Equation 3 also with a Tobit model. We furthermore control for household head and household characteristics  $(X_{ivt})$ , and include wave  $(\alpha_t)$  and village  $(\delta_t)$  fixed effects.  $u_{ivt}$  is the error term. Standard errors are clustered at the

 $<sup>^{6}\</sup>mathrm{In}$  contrast to other studies, we do not take net transfers. However, we address the robustness of using net transfers in Section 5.2.

rural household level.

We proxy household income with the annual household consumption net of transfers received and given.<sup>7</sup> Furthermore, the income variable is decomposed into permanent and transitory components. We roughly follow Kazianga (2006) and predict permanent income using characteristics that are permanent to the household (possession of livestock, assets, land owned, access to electricity, village fixed effects) and characteristics that determine transitory income (occurrence of negative shocks and wave fixed effects). The residual depicts the "unexplained" part of income (Paxson, 1992).

The income measure we use in our analysis causes some empirical problems. One concern is measurement error, which we nevertheless tried to limit by ensuring a thorough data collection. Enumerators were carefully trained and according to best practices, consumption was asked by categories to reduce recall error. Another concern is the endogeneity of transfers and income that can stem from reverse causality and omitted variables. To mitigate these biases, we propose a new instrumental variable (IV) (see Section 5.2). Apart from using an IV in some specifications, we also exploit the panel structure of our data allowing to include wave fixed effects and village fixed effects, which can at least control for all time-invariant factors at the village level. We are refraining from using household fixed effects in our preferred specification, because we are interested in the effect of some time-constant covariates on transfer patterns besides the income effect. The effect of these covariates, for instance, head of household characteristics and household composition, allows us to derive further implications for the targeting of public transfer initiatives. Yet, we test the robustness of our transfer derivatives when including household fixed effects and find them to be very similar. However, other omitted variables, in particular sender income and other sender characteristics, may still bias the results. The linked sample does allow us to control for the sender's income and therefore to check the validity of our results, although with a smaller number of observations (see Section 5.2).

As mentioned above, we include other household characteristics besides recipient income in the transfer function.  $X_{ivt}$  is a set of variables that potentially affect transfers received such as household composition and age of household head. A significant effect for older adults would, for instance, indicate an old-age support role of transfers (Cox, Galasso, & Jimenez, 2006). We moreover include a dummy indicating whether the household head is female, because many studies on private transfers find a positive correlation between female headship and transfers received (e.g., Cox et al., 2006, 2004; Cox & Jimenez, 1992; Kazianga, 2006; Lucas & Stark, 1985).

<sup>&</sup>lt;sup>7</sup>We take transfers received and given into account in order to reduce the endogeneity concern of taking household consumption as a proxy for household income. Household consumption is regularly used as a proxy for household income (Deaton, 1995). We find rural household consumption to be correlated with other income proxies such as livestock possession, land owned and the asset index. By using household consumption, we implicitly implicate the absence of access to credit and savings. Indeed, in a question on what measures rural households apply to cope with shocks (besides transfers), a share of below 0.1% reported using formal credit. A share of 2.5% reported using informal credit (from family or friends) as another measure and only 9.4% reported using savings.

Besides old-age and female support, transfers can also be explicitly motivated by support for chronically ill or in response to a shock, which is why we control for having a chronically sick person in the household and whether or not the household experienced a negative income shock in the past 12 months. Lastly, we also include the household head's level of education, where multiple interpretations have been put forward. On the one hand, it is correlated with lifetime resources, which may predict a negative effect on transfers. On the other hand, one may also expect a positive effect if it is correlated with past transfers, and thereby picks up the strength of parental altruism. A similar effect may show if education proxies for the recipient's ability to reciprocate (Cox & Fafchamps, 2007).

### 5 Results

#### 5.1 Main results: Transfer response to income

Table 4 reports the results of the estimated transfer functions. The observed patterns are used to learn about potential crowding-out effects ex-ante. In Columns 1, 3 and 5 we show the results of pooled OLS regressions. These ignore the censored nature of the dependent variable. To take this into account, we additionally use a Tobit model; the results are shown in Columns 2, 4 and 6. Addressing the non-linear transfer response, we first introduce income in exponential form (Columns 1 and 2). In Columns 3 and 4 we present the results of the continued spline specification as outlined in Equation 3 and in Columns 5 and 6 we further differentiate between permanent and transitory income splines.

The results shown in Columns 1 and 2 suggest a negative relationship between transfers received and the recipient's income. The coefficient estimates are small in size and even though the linear and squared income term are statistically significant, they are close to zero and economically negligible. Comparing the results of all specifications for the pooled OLS and Tobit estimations suggests that ignoring the censoring underestimates the effect associated with the recipient's income. Therefore, we focus our remarks on the results obtained by the Tobit model. Allowing the non-linear transfer response, by introducing continuous linear income quintile splines (see Columns 3 and 4), shows that the exponential form is a poor approximation of the relationship between transfers and income. The income splines are listed from lowest (poorest) to highest (richest). We still see a negative relationship between transfers and income, but only for the first and third income quintile. Particularly for the poorest households, this negative response is highly significant and sizeable. For this group, a 1,000 CFA Franc increase in income comes with a reduction in transfers by 1,170 CFA Franc. In contrast, for households in the second income quintile, a 1,000 CFA Franc increase in income comes with an increase in transfers by 450 CFA Franc. While the third income quintile displays again a negative relationship, the negative effect size is much smaller in magnitude to that of the first income group. We do not observe a significant relationship for the upper two income classes. The results of the F-Test indicate that the responses of the five income categories differ significantly from each other.

Figure 2 illustrates the predicted transfer response for households in each income quintile. The response is based on the Tobit estimates presented in Column 4 of Table 4. The figure shows that the poorest households are the ones who depend the most on private transfers and due to their financial position, would in principle be the main target group of public transfer programs. The strong negative response to a rise in income among this group indicates that public transfers would overproportionately offset their redistributive intention and actually lead to a decrease in overall available income. However, the figures presented here disguise an important aspect, namely that there is important heterogeneity with respect to the exposure to informal transfers in this group. Only about half of the households in the poorest income quintile are actual transfer recipients. Thus, for those that do not participate in transfer relations, public transfers would have a positive (overall) effect on income. For those that received transfer, however, the results suggest that public transfers would crowd out private ones.

Figure 2: Predicted transfer response to recipient income, rural households



*Note:* The transfer response is predicted using the Tobit estimation in Table 4, Column 4, and holding all other regressors fixed at their sample means. The vertical gray lines delimit the income quintiles.

In line with the conceptual framework laid out in Section 2, the negative relationship between transfers and income for the lowest income class is consistent with transfers being altruistically

	Annual transfers received $(1,000 \text{ CFA F})$					
	(1) pOLS	(2) Tobit	(3) pOLS	(4) Tobit	(5) pOLS	(6) Tobit
Annual HH income (1,000 CFA F)	-0.022*	-0.040*				
Annual HH income sq. $(/1,000)$	(0.013) $0.001^{*}$ (0.001)	(0.023) $0.002^{*}$ (0.001)				
1 <sup>st</sup> income spline	(0.001)	(0.001)	-1.065***	-1.167***	0.011	0.093
2 <sup>nd</sup> income spline			(0.071) $0.444^{***}$ (0.047)	(0.060) $0.448^{***}$ (0.073)	(0.027) 0.051 (0.057)	(0.077) 0.026 (0.144)
3 <sup>rd</sup> income spline			-0.102***	-0.136**	0.112	(0.144) $0.285^{*}$
4 <sup>th</sup> income spline			$(0.025) \\ 0.008$	$(0.057) \\ 0.018$	$(0.073) \\ 0.038$	$(0.163) \\ 0.039$
5 <sup>th</sup> income enline			(0.012)	(0.028)	(0.068)	(0.137)
5 income spine			(0.004)	(0.005)	(0.037)	(0.076)
Negative transitory income spline					$-0.443^{**}$	$-0.975^{***}$
Positive transitory income spline					(0.183) - $0.043^*$	-0.186***
Unexplained income $(1\ 000\ CFA\ F)$					(0.025)	(0.071)
(1,000 CFA F)					(0.005)	(0.012)
No. children 0-5	-2.0	-3.3	-1.9	-2.4	-2.7	-5.2
No. children 6-18	(2.3) -1.8	(4.6) -2.6	(2.0) -0.4	(3.6) -0.6	(2.3) -3.5*	(4.6) -6.8*
	(1.8)	(3.4)	(16.0)	(2.7)	(1.9)	(3.6)
No. adults 19-64	7.8**	10.6*	8.1***	10.5**	4.3	5.0
No. older odulto 65 -	(3.8)	(6.2)	(28.0)	(4.4)	(3.3)	(5.6)
No. older adults 05+	(3.4)	(6.7)	(25.0)	(4.8)	3.4	8.9 (6.7)
Children/adult_ratio	3.5	-0.6	(20.0)	1.0	(3.4)	3.2
Children/ adult Tablo	(5.0)	(8.2)	(3.1)	(5.4)	(4.9)	(8.2)
Age HH head (years)	-0.3	0.4	-0.8	-0.6	0.1	1.2
6 (8 )	(0.8)	(1.5)	(0.6)	(1.2)	(0.8)	(1.6)
Age HH head sq.	0.005	0.007	0.007	0.010	0.002	0.001
	(0.007)	(0.013)	(0.006)	(0.011)	(0.007)	(0.014)
HH head primary school $(=1)$	4.1	16.9	4.7	$13.6^{*}$	-0.7	6.0
	(4.9)	(11.4)	(3.6)	(8.0)	(4.7)	(11.1)
HH head secondary school $(=1)$	$17.5^{**}$	36.3**	12.8**	$24.5^{+++}$	12.6	$23.5^{*}$
Formale HH head $(-1)$	(0.7) 12.3	(10.4) 46 7***	(0.0)	(8.9)	(0.1) 10.7**	(14.1 <i>)</i> 65.1***
Temale IIII lieau $(-1)$	(8.3)	(14.9)	(6.2)	(10.9)	(8.4)	(16.0)
Chronically sick HH member $(=1)$	3.3	17.6**	2.4	13.1**	-2.1	3.7
	(4.2)	(8.4)	(3.0)	(5.9)	(4.8)	(9.8)
Shock in past 12 month $(=1)$	19.6***	66.3***	$13.6^{***}$	45.5***		
	(3.7)	(8.9)	(2.8)	(5.7)		
Constant	2.1	-183.8***	178.6***	70.8**	-22.9	-196.2***
	(19.7)	(46.4)	(20.3)	(34.7)	(23.2)	(59.2)
Village FE	YES	YES	YES	YES	YES	YES
Wave FE	YES	YES	YES	YES	YES	YES
Observations	3,752	3,752	3,752	3,752	3,752	3,752
No. of HHs Descreted	1,876	1,876	1,876	1,876	1,876	1,876
n-squarea F-Test p-value (quintiles)	0.040		0.024 0.000	0.000	0.035	0.007
F-Test p-value (positive/negative)			0.000	0.000	0.003	0.001

Table 4: Transfer response to recipient income, rural households

Note: Data from first and second wave. Robust standard errors in parentheses; standard errors are clustered at the household level. Transfers received include all monetary and in-kind transfers that a household received in the 12 months preceding the two survey waves; all in-kind transfers are assigned monetary values. Transfers received from various senders are aggregated. Annual (net) income is approximated with annual household consumption net of transfers received and given. The reference category for the household head's education is "no schooling". A shock is defined as an event that caused, according to the household, serious problems for the living conditions of the household. The reference are aggregated of the coefficients associated with positive and negative transitory income components. \*\*\* indicates significance at 1%, \*\* at 5%, and \* at 10%.

motivated. While the second income group depicts a positive relationship, the relationship between transfers and income turns negative again for the third income group. This pattern comes close to the inverted U-shaped relationship predicted by Cox (1987) and illustrated in Cox et al. (2004). That is, after the threshold where transfers cease being altruistic at the margin, transfers first rise and then fall. Though, in our sample the U-shape relationship only prevails between the second and third income group, whereas in  $\cos(1987)$  and  $\cos et al. (2004)$  this relationship spans over all upper income classes. Yet, we may interpret the income level between the first and second income quintile as the threshold where transfers switch to exchange related transfers. We cannot, however, conclude that altruistic or exchange related preferences actually exist. To further look into transfer motives, we can only draw upon qualitative evidence elicited from the urban households in our sample, which are often on the giving-side of the transfer relationship. Table 5 shows the share of urban households that agreed with different statements on sharing obligations. Almost all of the interviewed agree that, depending on the financial position, sharing is the norm (see Statements 1 and 2). In turn, Statements 3 and 4, which describe pure exchange relationships, meet little agreement. This could suggest that transfers in Burkina Faso are rather altruistically motivated, which is also what we observe empirically for our lowest income group. Furthermore, Statements 5 to 7 give an indication of how deep the expectations on giving are rooted in the households' social and traditional norms. While transfers might be altruistically motivated in the sense that they are reduced once the financial position of the recipient improves, partly the giving is also enforced through social pressure and individual guilt (Grimm, Hartwig, & Lay, 2017; Platteau, 2014).

	Urba	n HHs
Share that agrees with the following statements:	(1) Wave I (%)	(2) Wave II (%)
1. It is my duty to give to others as soon as I can.	91.9	94.1
2. My generosity depends on the financial burden at the moment they ask for my support.	85.4	76.9
3. I give to get something back.	10.1	4.1
4. If others give to me, I must return a gift of equal size to them.	11.0	5.0
5. I feel guilty if I do not give.	75.6	82.4
6. To have respect from your family, you have to support them financially or give gifts.	45.2	44.3
7. Whenever I have money, my spouse or other family members (in- or outside the household) request a share.	39.9	37.6

Table 5: Attitudes towards sharing obligations, urban households

Yet, as described in Section 2, transfers related to altruism and risk-sharing are analytically very close, making their implications and empirical predictions very similar. We therefore probe more closely into the possibility of risk-sharing by returning to Table 4. Columns 1 to 4 include household shocks in the transfer response function. The results suggest that households that have experienced a negative shock in the 12 months prior to the survey receive significantly higher transfers than those that have not. Around half of the negative shocks experienced by households in our sample are health-related shocks (for details see also Bocoum, Grimm, & Hartwig, 2018). The results suggest that the transfers received also have an insurance element. To further explore this question, we distinguish between permanent and transitory income in Columns 5 and 6. Permanent income enters in form of quintile income splines and transitory income enters the transfer function in form of a positive and negative transitory income spline. We observe that an increase in 1,000 CFA Franc in (positive) transitory income is associated with a 190 CFA Franc decrease in transfers. A decrease in transitory income, that is, an increase in negative transitory income, is associated with an increase of 980 CFA Franc. Consequently, in line with the effect of the (negative) shock dummy, rural households receive transfers in case of a negative income shock. In the presence of transitory income, we do not observe a significant relationship between transfers received and permanent income. Assuming that our permanent and transitory income measures are reliable, this could have important implications for the effectiveness of public transfers. While transfer types that affect the transitory component of household income (e.g., formal insurance) may crowd out private transfers, measures aiming at increasing permanent income may not. Hence, this could suggest that we do not observe crowding-out for regularly designed public transfers such as conditional or unconditional cash transfers.

So far, we mainly focused on the relationship between transfers and income. We now turn to the influence of other factors influencing transfers. This may allow us to derive further implications for the targeting of public transfer programs. Going back to Table 4, Column 4 (our preferred specification), we obtain statistically significant positive effects for the number of adults in the household, the presence of household members with chronic illness, and the education of the household head. Surprisingly, we do not observe a significant effect for the presence of older adults nor for female headship, although the sign of the effects is as expected. This is in contrast to evidence from other studies and with the anthropological literature on Burkina Faso, which describes a complex system of sharing norms that prescribe support of the elderly and less fortunate (see Fiske, 1990; Hammond, 1966; Saul, 1981, 1983). Hence, in terms of targeting, no clear suggestions can be derived for our sample.

#### 5.2 Robustness checks

The results of the previous section suggest potential crowding-out effects for the poorest income group, particularly for formal instruments that aim at increasing transitory income. In this section, we test in various ways the robustness of these findings by addressing several potential econometric identification problems as briefly discussed in Section 4. We address two potential endogeneity concerns that remain within our transfer response function: reverse causality and omitted variable bias. Concerning the former, we propose a new instrument. To control for sender income as one important source of omitted variable bias, we use data from the linked sample. In addition, we test for the robustness of our observed non-linear transfer response shown in Columns 3 and 4 in Table 4 for alternative specifications. The results are presented in Table C.1 in Appendix C. We find very similar patterns for using quartile instead of quintile income splines, for including household instead of village fixed effects and for using net transfers instead of gross transfers received.

#### Addressing reverse causality: IV estimation

To test whether our main results are robust to a possible endogeneity bias caused by reverse causality, we instrument income and re-estimate Equation 3 with an instrumental variable approach. Kazianga (2006) follows the approach of Paxson (1992) and instruments income using long-run regional rainfall data. Given the limited variation of rainfall within the community of Ziniaré, this is not a feasible option in our case. Our approach consists of using the household head's surname as an instrument for income. The underlying idea is that families that share the same surname are likely to belong to the same family dynasty and therefore are likely to have common determinants of intergenerational wealth. That is, some families are wealthier than others due to the accumulation of land, livestock and even political influence throughout generations. Hence, the surname provides informational content on long-term income of the respective family dynasty. It does not capture any short-term fluctuations and, thus, just yields a local average treatment effect (LATE) for longer term income.

In the rural sample, among the 1,876 households we have a total of 127 surnames in the first wave. The most common surname is shared by 19% of the household heads, the second most common by 7.1% and the third most common by 6.9%. On the other hand, 40 out of the 127 surnames are unique and hence carried only by one household head in the sample. The respective share of household heads is 2.1%. Furthermore, 31 of the surnames are shared by at least 10 household heads. The numbers are very similar in the second wave and suggest that there is enough variation to exploit the surnames as an instrument.

For the instrument to be relevant, we need the surnames to be sufficiently correlated with income. The first stage equation takes on the following form:

$$\sum_{k}^{5} I_{ivt} = \pi S_{ivt} + \beta X_{ivt} + \alpha_t + \delta_v + v_{ivt}, \qquad (4)$$

which is estimated for each income quintile spline separately. Hence, we have to instrument for five endogenous regressors and end up with five first stages (i.e., one for each quintile spline). This furthermore rests on the assumption that our instrument performs well in predicting income within each income quintile.  $S_{ivt}$  represents the surname of household (head) *i* in village *v* at

time t. The instrument enters the first stages in form of an (unordered) categorical variable. Table 6 summarizes relevant first stage statistics for each endogenous regressor separately. We do not provide overall first stage statistics, because their validity rests on the linearity assumption, which is not given in this case. Due to our piecewise linear approach, the validity of the tests is only given within each income quintile spline. The statistics in Table 6 confirm that our first stages are strong. Next to the relevance criterion, our instrument also needs to meet the exclusion criterion. This means that the surname may only influence transfers received through the income channel. Namely, we need to exclude any direct relation between the surname and transfers received. A potential threat to this assumption is the existence of specific sharing norms within each family dynasty. However, we believe in line with the anthropological literature that sharing norms are unlikely to be specific to a family but are rather shared on a higher level such as ethnicity, which in our case is predominantly the Mossi culture (Fiske, 1990). Hence, the surname should have an impact on transfers received only through the recipient's income but not on transfers directly. Yet, we acknowledge that the exclusion restriction in our case can be debated. For instance, we cannot rule out that certain dynasties are in a better position than others to build up a high-quality risk-sharing network and this may of course have a direct impact on transfers paid. A similar issue arises in those cases where the donor and recipient share the same surname. We therefore see the IV estimate rather as a robustness check which may give us another bound, but not necessarily as an estimator that we prefer over our estimates above. Finding an IV for income is in general far from obvious and even using rainfall data, where this is possible, may suffer from selective migration and endogenous adaption to severe weather conditions.

	Income splines				
	$(1) \\ 1^{\rm st}$	$\binom{(2)}{2^{\mathrm{nd}}}$	(3) $3^{\rm rd}$	$\begin{array}{c} (4) \\ 4^{\mathrm{th}} \end{array}$	$(5) \\ 5^{\rm th}$
Shea R-squared	0.018	0.032	0.032	0.041	0.016
1 <sup>st</sup> stage F-statistic p-value	$780.6 \\ 0.000$	$1,059.2 \\ 0.000$	$410.8 \\ 0.000$	$536.1 \\ 0.000$	417,217.0 0.000

Table 6: First-stage statistics for the income quintile splines

Column 2 of Table 7 presents the results of the IV estimation. In Column 1, we also include the pooled OLS estimates of Table 4, Column 3, for comparison. For the sake of simplicity, we abstract from zero censoring in the IV approach and only test the robustness of our pooled OLS results from Section 5.1. It can be seen that the pattern observed in Table 4 remains; we still find a strong negative significant response to an increase in household income among the poorest households. The effect size when instrumenting for income is on average lower (in absolute terms), which means that we slightly overestimate the transfer response to income for the poorest income quintile when failing to adjust for the endogeneity bias.

	Annual transfers receive (1,000 CFA F)		
	(1)	(2)	
	pOLS	ĪV	
1 <sup>st</sup> income spline	-1.065***	-1.000***	
	(0.071)	(0.159)	
2 <sup>nd</sup> income spline	$0.444^{***}$	$0.537^{***}$	
	(0.047)	(0.173)	
3 <sup>rd</sup> income spline	-0.102***	-0.116	
	(0.025)	(0.131)	
4 <sup>th</sup> income spline	0.008	0.047	
	(0.012)	(0.056)	
5 <sup>th</sup> income spline	0.004	0.003	
	(0.003)	(0.012)	
Recipient HH controls	YES	YES	
Village FE	YES	YES	
Wave FE	YES	YES	
Observations	3,752	3,752	
No. of HHs	1,876	1,876	
R-squared	0.524	0.515	

Table 7: IV-estimation of transfer response to recipient income, rural households

*Note:* Data from first and second wave. Robust standard errors in parentheses; standard errors are clustered at the household level. Transfers received include all monetary and in-kind transfers that a household received in the 12 months preceding the two survey waves; all in-kind transfers are assigned monetary values. Transfers received from various senders are aggregated. Annual (net) income is approximated with annual household consumption net of transfers received and given. Income is instrumented with the rural household head's surname. \*\*\* indicates significance at 1%, \*\* at 5%, and \* at 10%.

#### Addressing omitted variable bias: Controlling for sender income

In a next step, we use the linked sample to account for omitted variable bias, which may occur when failing to control for sender income in the transfer function. We do not have the information on sender incomes for the complete rural sample. This information is only available for the sub-sample of rural households for which we could also interview the urban households linked to them. This implies that this sub-sample is not necessarily representative for the total sample, yet, it can still serve to get an idea of the likely bias that occurs if the sender's income is not controlled and hence can serve as a sensitivity check of the robustness of the results in Section  $5.1.^8$  For the linked sample, Equation 3 is adjusted by including sender household income in

<sup>&</sup>lt;sup>8</sup>We test for systematic differences between the linked rural sample and the full rural sample in Appendix D. This comparison shows that rural households of the linked sample own slightly less livestock, but in turn have higher land ownership. They are also more likely to have a chronically sick household member and to have experienced a shock in the 12 months prior to the respective survey. The latter is not surprising, because only those that have received a transfer - which is often a response to a negative shock - can end up in the linked sample. For this reason, we believe that our linked sample is at least

form of a linear and quadratic term. The dependent variable is restricted to only covering the transfer received by the rural household by the linked urban household. However, we control for whether the rural household has transfer relations with other households outside the linked sample and also whether the linked urban household gave transfers to other than the linked rural household in form of dummies. We moreover include urban household and household head characteristics.

Columns 1 and 2 in Table 8 show the transfer response to recipient income without controlling for sender income. In Columns 3 and 4 we control for sender income, and in Columns 5 and 6 we add further sender characteristics. The comparison of the results of Column 2 with Columns 4 and 6 suggests that failing to control for the sender's income slightly overestimates the transfer response at the lowest tail of the income distribution. The potential bias, however, is very small. This is also what Altonji et al. (1997); Cox and Jimenez (1998); Cox and Rank (1992) and Kazianga (2006) suggested previously. However, they were not able to test this empirically.<sup>9</sup> The bias suggests that poor recipients are supported by slightly richer donors compared to somewhat less poor recipients.

The point estimate of the coefficient associated with sender income (Columns 4 and 6 in Table 8) is positive and statistically significant. When the sender's income increases by 1,000 CFA Franc, transfers received increase only marginally by 16 CFA Franc with a decreasing rate as the sender's income increases. Together with the negative relationship between recipient income and transfers, which we already observed in the larger sample and which also holds in the sub-sample here, the observed pattern is again indicative of altruism or risk-sharing as the underlying transfer motive – at least for the poorest households. Moreover, for the linked sub-sample we can check whether the rural households actually transfer back. We find that rural households only report very few transfers to their linked urban counterparts. In the first wave, only 3% of the rural households report a transfer to the linked urban household. In the second wave, this share stands at 5%. Hence, immediate exchange does not seem to be the dominant motive in our context. Yet, we cannot exclude the possibility of deferred or generalized reciprocity.

# 6 Conclusion

In this study, we analyze transfer responses to income in a poor rural setting in sub-Saharan Africa. While we also investigate transfer motives in the course of our empirical analysis, we use the motives as a means to assess ex-ante the potential of crowding-out effects for private transfers when introducing formal safety nets. Our approach follows to a large extent those of other studies in the developing country context and considers a non-linear transfer response.

comparable to those households in the full sample that also received a transfer.

<sup>&</sup>lt;sup>9</sup>We follow Kazianga (2006) and use Monte Carlo simulations to check how sensitive the bias is under different covariance levels between the recipient's and the sender's income. See Appendix E for details.

	Annual transfers received from linked HH $(1,000 \text{ CFA F})$					
	(1)	(2)	(3)	(4)	(5)	(6)
	pOLS	Tobit	pOLS	Tobit	pOLS	Tobit
1 <sup>st</sup> income spline	-0.574***	-0.611***	-0.560***	-0.592***	-0.555***	-0.586***
1	(0.148)	(0.150)	(0.152)	(0.156)	(0.153)	(0.156)
2 <sup>nd</sup> income spline	$0.175^{*}$	0.137	$0.180^{*}$	0.155	0.160	0.122
-	(0.106)	(0.137)	(0.107)	(0.139)	(0.107)	(0.136)
3 <sup>rd</sup> income spline	-0.020	0.009	-0.034	-0.020	-0.032	-0.006
	(0.072)	(0.091)	(0.075)	(0.094)	(0.074)	(0.092)
4 <sup>th</sup> income spline	0.041	0.023	0.041	0.025	0.044	0.027
	(0.058)	(0.072)	(0.057)	(0.070)	(0.057)	(0.069)
$5^{\rm th}$ income spline	0.006	0.015	0.004	0.013	0.002	0.011
	(0.018)	(0.020)	(0.017)	(0.020)	(0.018)	(0.020)
Ann. sending HH inc. $(1,000 \text{ CFA F})$			$0.010^{**}$	$0.014^{**}$	$0.009^{**}$	$0.016^{***}$
			(0.004)	(0.006)	(0.004)	(0.006)
Ann. sending HH inc. sq. $(/1,000)$			-0.001**	-0.001**	-0.001**	-0.001***
			(0.000)	(0.000)	(0.000)	(0.000)
Sending HH size					1.2	1.0
					(1.3)	(1.7)
Age sending HH head (years)					0.1	0.2
					(1.1)	(1.5)
Age sending HH head sq.					-0.002	-0.005
					(0.010)	(0.014)
Sending HH head prim. school $(=1)$					9.5	15.0
					(10.9)	(13.3)
Sending HH head second. school $(=1)$					5.4	7.3
					(6.6)	(8.5)
Female sending HH head $(=1)$					-14.6	-24.0
					(13.6)	(18.9)
Sending HH head is related $(=1)$					21.0**	71.9***
			i e e estatu		(8.3)	(19.6)
Constant	129.2**	99.7	108.7**	67.1	86.8	0.9
	(52.7)	(64.0)	(52.8)	(65.1)	(59.8)	(82.1)
Recipient HH controls	YES	YES	YES	YES	YES	YES
Village FE	YES	YES	YES	YES	YES	YES
Wave FE	YES	YES	YES	YES	YES	YES
Observations	580	580	580	580	580	580
No. of HHs	293	293	293	293	293	293
R-squared	0.270		0.281		0.289	
F-Test p-value (quintiles)	0.003	0.001	0.006	0.002	0.005	0.002

Table 8: Transfer response to income controlling for sender income, linked sample

*Note:* Data from first and second wave. Robust standard errors in parentheses; standard errors are clustered at the household level. Transfers received include all monetary and in-kind transfers that a household received from the linked urban household in the 12 months preceding the two survey waves; all in-kind transfers are assigned monetary values. Multiple transfers within one wave received from the linked urban household are aggregated. Annual (net) income is approximated with annual household consumption net of transfers received and given. The reference category for the household head's education is "no schooling". With respect to the transfer recipient and sender relation, the reference category to is "sending HH head is not related". Besides the recipient household controls (see Table 4), we also control for whether the household received transfer(s) from persons other than the linked urban household in the 12 months preceding the two survey waves. Moreover, we control for whether the linked urban household gave transfer(s) to other than the linked ruran household in the 12 months preceding the two survey waves. The F-Test p-values result from tests of equality of the effects across the income quintile splines. \*\*\* indicates significance at 1%, \*\* at 5%, and \* at 10%

We enrich the analysis by using novel data, especially data collected at both ends of the transfer relationship, and by using an innovative IV strategy.

Our results show that transfers overall decrease with recipient income and that this relationship is highly non-linear. The transfer response to income is very pronounced and negative at the low end of the income distribution (i.e., for the poorest of the poor). This is an already well-established finding. For example, Cox et al. (2004) find a similar pattern in the context of the Philippines and Maitra and Ray (2003) observe such replacement effects also in South Africa looking at public pensions to the poor. For the second income group, in turn, we observe a positive transfer derivative, which turns negative again for the third income group. This relationship, resembling the inverted U-shaped relationship predicted by Cox (1987), can be interpreted as the threshold where transfers cease being altruistic at the margin and switch to exchange related transfers. Despite the supporting qualitative evidence on sharing norms, we cannot, however, conclude from these findings that altruistic or exchange related preferences actually exist. Moreover, the observed transfer response at the lowest income group is also in line with predictions from risk-sharing models.

We show that our results regarding the relationship between recipient income and transfer size are robust to various alternative specifications and to the inclusion of sender income as a control variable, a potentially omitted variable most previous studies failed to control for due to data limitations. Furthermore, we propose a novel instrument for the recipient's income. While the existing literature uses rainfall as an instrument, which was not an option in our case given the limited geographical scope, we use family surnames as an instrument. We show that this instrument is relevant and also provide arguments that support the exclusion restriction. Even if not immune to criticism, we believe that it serves at least as a reasonable robustness check. The IV estimates suggest that our main results are robust with respect to the potential endogeneity of recipient income. In fact, not addressing endogeneity seems to lead to a slight overestimation of the negative transfer response among the poorest households.

Overall our results imply that the elasticity of transfers to recipient income is strongest for the poorest households, suggesting that the introduction of formal social protection would probably come with an overproportional crowding-out of private transfers. This is an important finding for the case of Burkina Faso, but also for many other poor rural settings within sub-Saharan Africa, as many countries in that region experiment, or at least think about the introduction of social protection policies. Yet, to a large extent, we find transfers to react particularly strong to an increase in transitory income. This would suggest that formal instruments that aim at increasing permanent income such as (conditional) cash transfers may not induce crowding-out of private transfers or only to a smaller extent.

# References

- Altonji, J. G., Hayashi, F., & Kotlikoff, L. J. (1997). Parental Altruism and Inter Vivos Transfers: Theory and Evidence. *Journal of Political Economy*, 105(6), 1121– 1166.
- Attanasio, O., & Ríos-Rull, J.-V. (2000). Consumption smoothing in island economies: Can public insurance reduce welfare? *European Economic Review*, 44(7), 1225– 1258.
- Barro, R. J. (1974). Are Government Bonds Net Wealth? Journal of Political Economy, 82(6), 1095–1117.
- Becker, G. S. (1974). A Theory of Social Interactions. Journal of Political Economy, 82(6), 1063–1093.
- Berg, A., Quirion, P., & Sultan, B. (2009). Weather-index drought insurance in Burkina-Faso: Assessment of its potential interest to farmers. Weather, Climate, and Society, 1(1), 71–84.
- Bernheim, B. D., Shleifer, A., & Summers, L. H. (1985). The Strategic Bequest Motive. Journal of Political Economy, 93(6), 1045–1076.
- Bocoum, F. Y., Grimm, M., & Hartwig, R. (2018). The health care burden in rural Burkina Faso: Consequences and implications for insurance design. SSM - Population Health, 6, 309–316.
- Cecchi, F., Duchoslav, J., & Bulte, E. (2016). Formal Insurance and the Dynamics of Social Capital: Experimental Evidence from Uganda. *Journal of African Economies*, 25(3), 418–438.
- Cochrane, J. H. (1991). A Simple Test of Consumption Insurance. Journal of Political Economy, 99(5), 957–976.
- Comola, M., & Fafchamps, M. (2017). The Missing Transfers: Estimating Misreporting in Dyadic Data. *Economic Development and Cultural Change*, 65(3), 549–582.
- Cox, D. (1987). Motives for Private Income Transfers. Journal of Political Economy, 95(3), 508–546.
- Cox, D., & Fafchamps, M. (2007). Extended Family and Kinship Networks: Economic Insights and Evolutionary Directions. In T. P. Schultz & J. Strauss (Eds.), *Handbook* of Development Economics (1st ed., Vol. 4, pp. 3711–3784). North Holland.
- Cox, D., Galasso, E., & Jimenez, E. (2006). Private Transfers in a Cross Section of Developing Countries (Center for Retirement Research Working Papers, No. CRR WP 2006-2). Chestnut Hill, MA: Center for Retirement Research, Boston College.
- Cox, D., Hansen, B. E., & Jimenez, E. (2004). How responsive are private transfers

to income? Evidence from a laissez-faire economy. *Journal of Public Economics*, 88(9-10), 2193–2219.

- Cox, D., & Jakubson, G. (1995). The connection between public transfers and private interfamily transfers. *Journal of Public Economics*, 57(1), 129–167.
- Cox, D., & Jimenez, E. (1992). Social Security and Private Transfers in Developing Countries: The Case of Peru. World Bank Economic Review, 6(1), 155–169.
- Cox, D., & Jimenez, E. (1998). Risk Sharing and Private Transfers: What about Urban Households? *Economic Development and Cultural Change*, 46(3), 621–637.
- Cox, D., & Rank, M. R. (1992). Inter-Vivos Transfers and Intergenerational Exchange. *Review of Economics and Statistics*, 74(2), 305–314.
- De Weerdt, J., Genicot, G., & Mesnard, A. (2019). Asymmetry of Information within Family Networks. *Journal of Human Resources*, 54(1), 225–254.
- Deaton, A. (1995). Data and econometric tools for development analysis. In J. Behrman & T. Srinivasan (Eds.), *Handbook of Development Economics* (Vol. 3, pp. 1785–1882). North Holland.
- Dercon, S., Hill, R. V., Clarke, D., Outes-Leon, I., & Seyoum Taffesse, A. (2014). Offering rainfall insurance to informal insurance groups: Evidence from a field experiment in Ethiopia. *Journal of Development Economics*, 106, 132–143.
- Dercon, S., & Krishnan, P. (2003). Risk sharing and public transfers. *Economic Journal*, 113(486), 86–94.
- Edin, K., & Lein, L. (1997). Making Ends Meet: How Single Mothers Survive Welfare and Low-Wage Work. New York, NY: Russell Sage Foundation.
- Fafchamps, M., & Gubert, F. (2007). The formation of risk sharing networks. Journal of Development Economics, 83(2), 326–350.
- Fafchamps, M., & Lund, S. (2003). Risk-sharing networks in rural Philippines. Journal of Development Economics, 71(2), 261–287.
- Fiske, A. P. (1990). Relativity within Moose ("Mossi") Culture: Four Incommensurable Models for Social Relationships. *Ethos*, 18(2), 180–204.
- Geng, X., Janssens, W., Kramer, B., & van der List, M. (2018). Health insurance, a friend in need? Impacts of formal insurance and crowding out of informal insurance. World Development, 111, 196–210.
- Grimm, M., Hartwig, R., & Lay, J. (2017). Does forced solidarity hamper investment in small and micro enterprises? *Journal of Comparative Economics*, 45(4), 827–846.
- Grimm, M., Wetta, C., & Nikiema, A. (2016). Burkina Faso: Shipping around the Malthusian trap. In C. Arndt, A. McKay, & F. Tarp (Eds.), Growth and Poverty in Sub-Saharan Africa (chap. 8). Oxford, UK: Oxford University Press.

- Hammond, P. B. (1966). Yatenga: Technology in the Culture of a West African Kingdom. New York, NY: Free Press.
- Heemskerk, M., Norton, A., & de Dehn, L. (2004). Does Public Welfare Crowd Out Informal Safety Nets? Ethnographic Evidence from Rural Latin America. World Development, 32(6), 941–955.
- Jensen, R. T. (2003). Do private transfers 'displace' the benefits of public transfers? Evidence from South Africa. *Journal of Public Economics*, 88(1-2), 89–112.
- Jimenez, E. V., & Brown, R. P. C. (2012). How Responsive are Remittances to Recipients' Welfare? Evidence from Fiji. International Migration, 51(S1), e179–e201.
- Kazianga, H. (2006). Motives for household private transfers in Burkina Faso. Journal of Development Economics, 79(1), 73–117.
- La Ferrara, E. (2003). Kin Groups and Reciprocity: A Model of Credit Transactions in Ghana. *American Economic Review*, 93(5), 1730–1751.
- La Ferrara, E. (2010). Family and Kinship Ties in Development: An Economist's Perspective. In J.-P. Platteau & R. Peccoud (Eds.), *Culture, Institutions, and Development: New Insights Into an Old Debate* (1st ed., pp. 125–142). London: Routledge.
- Landmann, A., Vollan, B., & Frölich, M. (2012). Insurance versus Savings for the Poor: Why One Should Offer Either Both or None (IZA Discussion Papers, No. 6298).
  Bonn: Institute of Labor Economics (IZA).
- Lenel, F., & Steiner, S. (2017). Insurance and Solidarity: Evidence from a Lab-in-the-Field Experiment in Cambodia (IZA Discussion Papers, No. 10986). Bonn: Institute of Labor Economics (IZA).
- Ligon, E., Thomas, J. P., & Worrall, T. (2002). Informal Insurance Arrangements with Limited Commitment: Theory and Evidence from Village Economies. *Review of Economic Studies*, 69(1), 209–244.
- Lin, W., Liu, Y., & Meng, J. (2014). The crowding-out effect of formal insurance on informal risk sharing: An experimental study. *Games and Economic Behavior*, 86, 184–211.
- Lucas, R. E. B., & Stark, O. (1985). Motivations to Remit: Evidence from Botswana. Journal of Political Economy, 93(5), 901–918.
- Mace, B. J. (1991). Full Insurance in the Presence of Aggregate Uncertainty. Journal of Political Economy, 99(5), 928–956.
- Maitra, P., & Ray, R. (2003). The effect of transfers on household expenditure patterns and poverty in South Africa. *Journal of Development Economics*, 71(1), 23–49.
- Mobarak, A. M., & Rosenzweig, M. R. (2013). Informal Risk Sharing, Index Insurance, and Risk Taking in Developing Countries. *American Economic Review: Papers &*

*Proceedings*, 103(3), 375-380.

- O'Connor, J. S., Orloff, A. S., & Shaver, S. (1999). States, Markets, Families: Gender, Liberalism and Social Policy in Australia, Canada, Great Britain, and the United States. Cambridge, UK: Cambridge University Press.
- Paxson, C. H. (1992). Using Weather Variability to Estimate the Response of Savings to Transitory Income in Thailand. American Economic Review, 82(1), 15–33.
- Platteau, J.-P. (2014). Redistributive Pressures in Sub-Saharan Africa: Causes, Consequences, and Coping Strategies. In E. Akyeampong, R. H. Bates, N. Nunn, & J. A. Robinson (Eds.), Africa's Development in Historical Perspective (pp. 153– 207). Cambridge: Cambridge University Press.
- Saul, M. (1981). Beer, sorghum and women: Production for the market in rural Upper Volta. Africa: Journal of the International African Institute, 51(3), 746–764.
- Saul, M. (1983). work parties, wages, and accumulation in a Voltaic village. American Ethnologist, 10(1), 77–96.
- Strupat, C., & Klohn, F. (2018). Crowding out of solidarity? Public health insurance versus informal transfer networks in Ghana. World Development, 104, 212–221.
- Townsend, R. M. (1994). Risk and Insurance in Village India. *Econometrica*, 62(3), 539–591.
- UNDP. (2019). Beyond income, beyond averages, beyond today: Inequalities in human development in the 21st century (Human Development Report 2019). New York, NY: United Nations Development Programme.
- World Bank. (2018). The State of Social Safety Nets 2018 (World Bank Report). Washington, D.C.: World Bank.

# Appendix

### **Appendix A: Attrition**

No. of HHs

		HH in both waves $(=1$	.)
	(1)	(2)	(3)
	Rural sample	Linked rural sample	Urban sample
Annual HH income (1,000 CFA F)	-0.000*	-0.000	0.000
	(0.000)	(0.000)	(0.000)
Livestock (TLU)	0.009	-0.026	
	(0.016)	(0.046)	
Land owned (ha)	0.090	0.016	
	(0.055)	(0.037)	
No. children 0-5	-0.178	0.002	$0.414^{**}$
	(0.119)	(0.145)	(0.204)
No. children 6-18	-0.018	0.121	-0.085
	(0.100)	(0.098)	(0.132)
No. adults 19-64	0.244*	0.064	-0.022
	(0.127)	(0.135)	(0.130)
No. older adults 65+	0.003	0.118	-1.013**
	(0.207)	(0.211)	(0.398)
Children/adult ratio	$0.399^{*}$	-0.037	0.098
7	(0.213)	(0.227)	(0.345)
Age HH head (years)	-0.001	0.001	0.022
0 (0 /	(0.008)	(0.010)	(0.014)
HH head primary school $(=1)$	-0.179	0.241	-0.161
1 0 ( /	(0.282)	(0.410)	(0.324)
HH head secondary school $(=1)$	0.005	-0.482	-0.382
	(0.274)	(0.295)	(0.286)
HH has business $(=1)$		()	-0.079
			(0.266)
Female HH head $(=1)$	-0.462	0.179	0.377
	(0.344)	(0.504)	(0.641)
Chronically sick HH member $(=1)$	0.405*	0.001	0.276
	(0.245)	(0.251)	(0.295)
Shock in past 12 month $(=1)$	0.183	-0.219	0.615**
r ( -)	(0.212)	(0.236)	(0.272)
Annual transfers received (1.000 CFA F)	0.002	0.000	-0.002
	(0.002)	(0.001)	(0.002)
Annual transfers given (1,000 CFA F)	-0.001	0.071**	-0.000
	(0.001)	(0.034)	(0.000)
Constant	1 559***	0.023	-0.498
	(0.515)	(0.625)	(0.581)
Observations	1 996	359	350

Table A.1: Test for systematic attrition between the first and second wave

1,996

359

359

Note: Data from first wave. Logit estimates. Robust t-statistics in parentheses. Transfers received and given include all monetary and in-kind transfers that a household has received or given in the 12 months preceding the survey wave; all in-kind transfers are assigned monetary values. Transfers received from various senders or given to various recipients are aggregated. Annual (net) income is approximated with annual household consumption net of transfers received and given. The value of livestock is expressed in tropical livestock units with the following conversion factor: cattle=0.7, sheep=0.1, goats=0.1, pigs=0.2 and chicken=0.01. The asset index combines housing materials and the possession of durable consumption goods; it is calculated using factor analysis. The reference category for the household head's education is "no schooling". A shock is defined as an event that caused, according to the household, serious problems for the living conditions of the household. \*\*\* indicates significance at 1%, \*\* at 5%, and \* at 10%.

#### **Appendix B: Transfer patterns**

Table B.1: Annual transfers received for each income quintile, rural households

	Rural HHs				
	(1)	(2)	(3)	(4)	(5)
		Incon	ne quir	ntiles	
	$1^{\mathrm{st}}$	$2^{nd}$	$3^{\rm rd}$	$4^{\rm th}$	$5^{\mathrm{th}}$
Wave I					
Received informal transfer $(\%)$	45.9	43.7	34.0	43.7	40.6
Annual transfers size $(1,000 \text{ CFA F})$	111.8	43.6	54.9	55.7	87.0
Wave II					
Received informal transfer $(\%)$	49.8	40.2	33.8	35.1	33.6
Annual transfers size $(1,000 \text{ CFA F})$	114.2	51.0	40.0	44.3	57.2

*Note:* Transfers received include all monetary and in-kind transfers that a household has received in the 12 months preceding the two survey waves; all in-kind transfers are assigned monetary values. Transfers received from various senders are aggregated. The mean annual transfer sizes within each income quintile are conditional on receiving a transfer. 1 USD corresponds to approx. 577 CFA F.

	Wave I		Wave II	
	(1)Transfer size (1,000 CFA F)	(2) No. of HHs	(3) Transfer size (1,000 CFA F)	(4) No. of HHs
Rural HHs; annual transfers received				
Total	72.1	785	63.1	712
Sender relation				
- Extended family	67.1	730	58.6	666
- Non-family	85.7	89	75.7	78
Sender location	24.0		91 0	<b>۲</b>
- Same village	34.6	85	31.8	85
- Other rural area	32.2	147	46.7	150
- Urban Abread	(2.1	010 E1	01.0 00.2	08
- Abroad Type	90.1	51	09.0	98
- Shock	81.6	273	96.1	180
Agricultural shock	64.3	46	41 7	20
Accident/illness	81 4	40 134	91.3	105
Death of family member	86.9	55	149.7	44
Asset shock (loss/damage)	89.4	18	63.3	7
Other	67.0	23	31.3	16
- Regular	58.6	586	44.6	600
Subsistence needs	48.9	416	41.2	450
Chronic illness expenses	58.4	49	26.9	15
Child education	59.6	15	89.0	15
Ceremony	23.3	45	18.0	61
Other	70.8	130	43.9	99
Rural HHs; annual transfers given				
Total	44.2	190	27.7	375
Recipient relation				
- Extended family	46.0	162	28.4	344
- Non-family	26.4	36	14.0	43
Recipient location	10.0	<u>م</u> ۲	21.2	07
- Same village	49.3	35	21.2	97 170
- Other rural area	38.3	83	22.3	179
- Urban Abroad	40.7	83 2	27.8	134 17
- Abroad Tupo	30.1	3	30.0	11
Subsistance needs	22.0	114	17.2	227
Chronic illness expenses	22.9	11	55.0	207 10
Accident/illness	125.9	12	28.3	21
Child education	59.5	16	74.8	$\frac{21}{25}$
Ceremony	28.1	8	11.4	48
Other	70.8	40	49.6	55
Total no. of HHs		1.876		1.876

Table B.2: Size of informal transfers received and given in the 12 months preceding the two survey waves (detailed version), rural households

*Note:* Transfers received and given include all monetary and in-kind transfers that a household has received or given in the 12 months preceding the two survey waves; all in-kind transfers are assigned monetary values. Transfers received from various senders or given to various recipients are aggregated. The mean values in Columns 1 and 3 are conditional on receiving/giving a transfer. 1 USD corresponds to approx. 577 CFA F. A shock is defined as an event that caused, according to the household, serious problems for the living conditions of the household.

	Wave I		Wave II	
	(1)Transfer size (1,000 CFA F)	(2) No. of HHs	(3) Transfer size (1,000 CFA F)	(4) No. of HHs
Urban HHs; annual transfers received				
Total	46.3	56	34.1	50
Sender relation				
- Extended family	107.0	41	115.0	36
- Non-family	98.0	22	92.2	17
Sender location				
- Home village	99.3	7	186.7	6
- Other rural area	40.0	6	129.0	3
- Urban	77.0	36	94.1	33
- Abroad	257.7	11	84.3	13
Type				
- Shock	99.7	26	131.2	13
Business shock	10.0	1	0.0	0
Accident/illness	77.4	19	47.5	8
Death of family member	282.5	2	1,000.0	1
Other	136.3	4	81.3	4
- Regular	109.8	36	95.3	42
Subsistence needs	97.3	19	53.2	26
Chronic illness expenses	52.5	2	226.7	3
Child education	57.5	2	200.0	2
Ceremony	306.7	3	62.5	2
Other	64.3	15	157.2	9
Urban HHs; annual transfers given				
Total	113.6	198	137.9	188
Recipient relation				
- Extended family	102.6	195	135.2	186
- Non-family	118.9	21	40.8	19
Recipient location				
- Home village	99.5	180	107.7	181
- Other rural area	60.5	28	157.4	14
- Urban	89.4	31	129.4	30
- Abroad	36.7	3	67.6	5
Type				
Subsistence needs	75.9	165	104.4	145
(Chronic) illness expenses	166.2	28	161.8	20
Child education	78.0	9	105.7	14
Ceremony	63.5	5	72.9	20
Other	110.2	39	148.7	31
Total no. of HHs		225		225

Table B.3: Size of informal transfers received and given in the 12 months preceding the two survey waves (detailed version), urban households

*Note:* Transfers received and given include all monetary and in-kind transfers that a household has received or given in the 12 months preceding the two survey waves; all in-kind transfers are assigned monetary values. Transfers received from various senders or given to various recipients are aggregated. The mean values in Columns 1 and 3 are conditional on receiving/giving a transfer. 1 USD corresponds to approx. 577 CFA F. A shock is defined as an event that caused, according to the household, serious problems for the living conditions of the household.

results
main
of
analysis
Sensitivity
ü
ppendix

, rural households	
specifications	
alternative	
sponse to	
f transfer re	
Robustness o	
Table C.1: I	

			Annual t	ransfers rec	eived $(1,000)$	CFA F)		
	Quintile	splines	Quartile	e splines	HH	FE	Net tre	ansfers
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
	pOLS	Tobit	pOLS	Tobit	pOLS	Tobit	pOLS	Tobit
1 <sup>st</sup> income spline	$-1.065^{***}$	$-1.167^{***}$	-0.944***	$-1.058^{***}$	$-1.070^{***}$	$-1.171^{***}$	$-1.051^{***}$	$-1.157^{***}$
	(0.071)	(0.060)	(0.092)	(0.078)	(0.090)	(0.073)	(0.074)	(0.065)
2 <sup>nd</sup> income spline	$0.444^{***}$	$0.448^{***}$	$0.340^{***}$	$0.319^{***}$	$0.375^{***}$	$0.414^{***}$	$0.430^{***}$	$0.430^{***}$
	(0.047)	(0.073)	(0.045)	(0.058)	(0.076)	(0.113)	(0.048)	(0.076)
$3^{rd}$ income spline	$-0.102^{***}$	$-0.136^{**}$	-0.053***	-0.038	$-0.094^{*}$	$-0.164^{*}$	$-0.108^{***}$	$-0.136^{**}$
	(0.025)	(0.057)	(0.013)	(0.028)	(0.053)	(0.092)	(0.025)	(0.058)
$4^{\rm th}$ income spline	0.008	0.018	0.004	0.005	0.004	0.009	-0.016	-0.005
	(0.012)	(0.028)	(0.003)	(0.005)	(0.026)	(0.047)	(0.013)	(0.029)
$5^{\rm th}$ income spline	0.004	0.005			0.002	0.004	0.002	0.005
	(0.003)	(0.005)			(0.004)	(0.009)	(0.003)	(0.005)
Recipient HH controls	YES	YES	$\mathbf{YES}$	$\mathbf{YES}$	YES	YES	YES	YES
Village FE	YES	$\mathbf{YES}$	$\mathbf{YES}$	$\mathbf{YES}$	NO	NO	YES	$\mathbf{YES}$
HH FE	ON	NO	NO	NO	$\mathbf{YES}$	$\mathbf{YES}$	NO	ON
Wave FE	YES	$\mathbf{YES}$	YES	$\mathbf{YES}$	$\mathbf{YES}$	$\mathbf{YES}$	$\mathbf{YES}$	YES
Observations	3,752	3,752	3,752	3,752	3,752	3,752	3,752	3,752
Number of households	1,876	1,876	1,876	1,876	1,876	1,876	1,876	1,876
R-squared	0.524		0.481		0.777		0.490	
F-Test p-value (quintiles)	0.000	0.000			0.000	0.000	0.000	0.000
F-Test p-value (quartiles)			0.000	0.000				
<i>Note:</i> Data from first and second w all monetary and in-kind transfers th Transfers received from various sende Further controls are equivalent to th	vave. Robust sta hat a household ers are aggregate nose in Table 4.	andard errors in received in the d. Annual (net) The F-Test p-va	parentheses; sta 12 months prece income is approx lues result from	andard errors an ding the two su cimated with an tests of equality	e clustered at th rvey waves; all ii nual household co v of the effects av	ae household lev n-kind transfers onsumption net e cross the income	el. Transfers rec are assigned moi of transfers receiv e quintile/quartil	ceived include netary values. red and given. e splines. ***
indicates significance at $1\%$ , ** at $5\%$	%, and * at 10%.							

#### Appendix D: The linked sample

	(1)
	Linked sample $(=1)$
Annual HH income (1,000 CFA F)	-0.000
	(0.000)
Livestock (TLU)	-0.064***
	(0.021)
Land owned (ha)	0.037**
	(0.017)
No. children 0-5	0.014
	(0.076)
No. children 6-18	0.028
	(0.057)
No. adults 19-64	-0.006
	(0.080)
No. older adults 65+	0.056
	(0.123)
Children/adult ratio	0.012
	(0.131)
Age HH head (years)	0.006
	(0.006)
HH head primary school $(=1)$	0.063
	(0.216)
HH head secondary school $(=1)$	0.220
	(0.180)
Female HH head $(=1)$	0.044
	(0.277)
Chronically sick HH member $(=1)$	$0.241^{*}$
	(0.145)
Shock in past 12 month $(=1)$	$0.530^{***}$
	(0.135)
Annual transfers received $(1,000 \text{ CFA F})$	$0.003^{***}$
	(0.001)
Annual transfers given $(1,000 \text{ CFA F})$	-0.002
	(0.003)
Constant	-2.528***
	(0.350)
Observations	1.876
No. of HHs	1,876

Table D.1: Test for systematic differences between rural and linked rural sample

Note: Data from first wave. Logit estimates. Robust t-statistics in parentheses. Transfers received and given include all monetary and in-kind transfers that a household has received or given in the 12 months preceding the survey wave; all in-kind transfers are assigned monetary values. Transfers received from various senders or given to various recipients are aggregated. Annual (net) income is approximated with annual household consumption net of transfers received and given. The value of livestock is expressed in tropical livestock units with the following conversion factor: cattle=0.7, sheep=0.1, goats=0.1, pigs=0.2 and chicken=0.01. The asset index combines housing materials and the possession of durable consumption goods; it is calculated using factor analysis. The reference category for the household head's education is "no schooling". A shock is defined as an event that caused, according to the household, serious problems for the living conditions of the household. \*\*\*\* indicates significance at 1%, \*\* at 5%, and \* at 10%.

#### Appendix E: Omitted variable bias

To explore whether our estimates are still affected by other unobservable factors that are correlated with both the recipient's and the sender's income, we follow Kazianga (2006) and use Monte Carlo simulations to check how sensitive the bias is under different covariance levels between the recipient's and the sender's income. The transfer function used in our data generating process is  $T_i = \alpha_0 + \alpha_1 I_{ri} + \alpha_2 I_{di} + \epsilon_i$  with  $\alpha_1 = -0.3$  and  $\alpha_2 = 0.3$ . We vary the covariance between the recipient's and the sender's income. In line with Kazianga (2006), we assume zero covariance and a covariance of 0.250, respectively. We also calculate the prevailing covariance between recipient and sender income in our linked sample, which is 0.141. Table E.1 reports the point estimates for different sample sizes, to vary the degree of precision, and estimations using OLS and fixed effects. Beneath the point estimates are the standard errors and the absolute deviation from the true value of -0.3. The simulations suggest that the bias is between 0.110 to 0.130 for an OLS and a fixed effects model, respectively. We see that the bias increases with the assumed covariance. For the covariance measured in our data, both models perform very similar.

	$\operatorname{Cov}(I_d,$	$I_r) = 0.000$	$\operatorname{Cov}(I_d, I_d)$	$I_r) = 0.141$	$\operatorname{Cov}(I_d, I_d)$	$I_r) = 0.250$
	(1)	(2)	(3)	(4)	(5)	(6)
Sample size	OLS	FÉ	OLS	FE	OLS	$\mathbf{FE}$
100	-0.292	-0.330	-0.185	-0.167	-0.065	-0.106
	(0.006)	(0.024)	(0.009)	(0.031)	(0.009)	(0.042)
	[0.008]	[-0.030]	[0.115]	[0.133]	[0.235]	[0.194]
1000	-0.299	-0.300	-0.176	-0.175	-0.080	-0.079
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
	[0.001]	[0.000]	[0.124]	[0.125]	[0.220]	[0.221]
10,000	-0.300	-0.300	-0.176	-0.176	-0.079	-0.079
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.009)
	[0.000]	[0.000]	[0.124]	[0.124]	[0.221]	[0.221]

Table E.1: Monte Carlo simulations to address omitted variable bias

*Note:* Monte Carlo simulations to check how sensitive the bias is under different covariance levels between the recipient's and the sender's income. We assume zero covariance, a covariance of 0.250 and we calculate the covariance between recipient and sender income in our (linked) sample (i.e., 0.141). The table shows the point estimates for different sample sizes and estimations using OLS and fixed effects. Beneath the point estimates are the standard errors (in parentheses) and the absolute deviation from the true value of -0.3 [in brackets].

# Volkswirtschaftliche Reihe der Passauer Diskussionspapiere

### **Bisher sind erschienen:**

V-1-98	Gerhard Rübel, Can adjustments to working hours help reduce unemployment?
V-2-98	Martin Werding, Pay-as-you-go Public Pension Schemes and Endogenous Fertility: The Reconstruction of Intergenerational Exchange
V-3-98	Carsten Eckel, International Trade, Direct Investment, and the Skill Differential in General Equilibrium
V-4-98	Reinar Lüdeke, Das Staatsbudget und intergenerationelle Umverteilung, Das Staatsvermögen als Instrument intergenerativer Verteilungspolitik und der "generational accounting"-Ansatz: Alter Wein in neuen (höherwertigen) Schläuchen?
V-5-98	Anja Klüver und Gerhard Rübel, Räumliche Industriekonzentration und die kom- parativen Vorteile von Ländern - eine empirische Studie der Europäischen Union
V-6-98	Klaus Beckmann und Elisabeth Lackner, Vom Leviathan und von optimalen Steuern
V-7-98	Martin Werding, The Pay-as-you-go Mechanism as Human Capital Funding: The "Mackenroth hypothesis" Revisited
V-8-98	Reinar Lüdeke und Klaus Beckmann, Social Costs of Higher Education: Production and Financing. The Case of Germany (1994)
V-9-98	Gerhard Rübel, "Faire" Löhne und die Flexibilität von Arbeitsmärkten in einem Zwei-Sektoren-Modell
V-10-98	Klaus Beckmann, Notizen zum Steueranteil von Rentenversicherungsbeiträgen im Umlageverfahren
V-11-98	Christian Jasperneite und Hans Joachim Allinger, Trendwende am westdeutschen Arbeitsmarkt? - Eine ökonometrische Analyse
V-12-98	Christian Jasperneite und Hans Joachim Allinger, Langfristige Perspektiven für den westdeutschen Arbeitsmarkt: Was sagen die Gesetze von Okun und Verdoorn?
V-13-98	Hans Joachim Allinger und Christian Jasperneite, Saisonbereinigung von Arbeitsmarktdaten bei aktiver Arbeitsmarktpolitik
V-14-99	Reinar Lüdeke und Klaus Beckmann, Hochschulbildung, Humankapital und Beruf: Auswertung einer Längsschnittsbefragung Passauer Absolventen 1988 - 1998

V-15-99	Gerhard Rübel, Volkseinkommenssteigerung durch ausgabenfinanzierte Steuersenkung - Eine Umkehrung des Haavelmo-Theorems für offene Volkswirt- schaften
V-16-99	Silke Klüver, Konzentrationsursachen in der europäichen Versicherungsbranche - eine empirische Untersuchung
V-17-99	Reinar Lüdeke, Familienlastenausgleich, Elternleistungsausgleich und die Neu- fundierung der umlagefinanzierten Altersversorgung
V-18-99	Anja Klüver und Gerhard Rübel, Industrielle Konzentration als Kriterium für die Geeignetheit eines einheitlichen Währungsraums – Eine empirische Untersuchung der Europäischen Union von 1972 bis 1996
V-19-00	Carsten, Eckel, Fragmentation, Efficiency-seeking FDI, and Employment
V-20-00	Christian Jasperneite, Understanding Hysteresis in Unemployment: The German Case
V-21-00	Jörg Althammer, Reforming Family Taxation
V-22-00	Carsten Eckel, Labor Market Adjustments to Globalization: Unemployment versus Relative Wages
V-23-00	Klaus Beckmann, Tax Competition through Tax Evasion
V-24-01	Klaus Beckmann, Steuerhinterziehung, begrenzte Rationalität und Referenzab- hängigkeit: Theorie und experimentelle Evidenz
V-25-01	Klaus Beckmann, Solidarity, Democracy, and Tax Evasion: an Experimental Study
V-26-04	Michael Fritsch, Udo Brixy und Oliver Falck, The Effect of Industry, Region and Time on New Business Survival - A Multi-Dimensional Analysis
V-27-04	Gerhard D. Kleinhenz, Bevölkerung und Wachstum - Die Bevölkerungs- entwicklung in Deutschland als Herausforderung für Wirtschafts- und Sozialpolitik
V-28-04	Johann Graf Lambsdorf, The Puzzle with Increasing Money Demand - Evidence from a Cross-Section of Countries
V-29-04	Frauke David, Oliver Falck, Stephan Heblich und Christoph Kneiding, Generationsgerechtigkeit und Unternehmen
V-30-04	Roland Engels <sup>†</sup> , Zur mikroökonomischen Fundierung der Geldnachfrage in allgemeinen Gleichgewichtsmodellen

V-31-05	Johann Graf Lambsdorff, Between Two Evils – Investors Prefer Grand Corruption!
V-32-05	Oliver Falck, Das Scheitern junger Betriebe – Ein Überlebensdauermodell auf Basis des IAB-Betriebspanels
V-33-05	Raphaela Seubert - On the Nature of the Corrupt Firm: Where to Situate Liability?
V-34-05	Johann Graf Lambsdorff – Consequences and Causes of Corruption – What do We Know from a Cross-Section of Countries?
V-35-05	Stephan Heblich - Arbeitszeitflexibilisierung Revisited
V-36-05	Oliver Falck und Stephan Heblich - Das Konzept der eigenverantwortlichen Generation zur Bewältigung des demographischen Wandels
V-37-05	Florian Birkenfeld, Daniel Gastl, Stephan Heblich, Ferry Lienert, Mascha Maergoyz, Oksana Mont und Andrius Plepys - Product ban versus risk management by setting emission and technology requirements – the effect of different regulatory schemes taking the use of trichloroethylene in Sweden and Germany as an example
V-38-05	Johann Graf Lambsdorff - Determining Trends for Perceived Levels of Corruption
V-39-05	Oliver Falck - Mayflies and Long-Distance Runners: The Effects of New Business Formation on Industry Growth
V-40-05	Johann Graf Lambsdorff und Christian Engelen - Hares and Stags in Argentinean Debt Restructuring
V-41-05	Johann Graf Lambsdorff und Mathias Nell – Let Them Take Gifts, and Cheat Those Who Seek Influence
V-42-06	Hans Joachim Allinger – Bürgerversicherung und Kopfpauschale haben vieles gemeinsam – Anmerkungen zur Diskussion einer Reform der gesetzlichen Krankenversicherung
V-43-06	Michael Schinke und Johann Graf Lambsdorff - Insider Trading among Central Bankers – a Treatise on Temptation and Policy Choice
V-44-06	Johann Graf Lambsdorff und Hady Fink - Combating Corruption in Colombia: Perceptions and Achievements
V-45-06	Oliver Falck und Stephan Heblich - Corporate Social Responsibility: Einbettung des Unternehmens in das Wirtschaftssystem
V-46-06	Johann Graf Lambsdorff und Luka Bajec - There Is No Bank Lending Channel!
V-47-06	Christian Engelen und Johann Graf Lambsdorff - Das Keynesianische Konsensmodell

V-48-07	Stephan Heblich - Eigenverantwortliche Individuen und Pro-Aktive Unternehmen
V-49-07	Christian Engelen und Johann Graf Lambsdorff - Das Keynesianische Konsensmodell einer offenen Volkswirtschaft
V-50-07	Christian Engelen und Johann Graf Lambsdorff - Fairness in Sovereign Debt Restructuring
V-51-07	Johann Graf Lambsdorff und Björn Frank - Corrupt Reciprocity - an Experiment
V-52-07	Mathias Nell - Strategic Aspects of Voluntary Disclosure Programs for Corruption Offences - Towards a Design of Good Practice -
V-53-07	Mathias Nell - Contracts Induced by Means of Bribery - Should they be Void or Valid?
V-54-08	Michael Pflüger – Die Neue Ökonomische Geographie: Ein Überblick
V-55-08	Florian Birkenfeld und Shima'a Hanafy - Wie zentral sind die Abschlussprüfungen an deutschen Schulen wirklich?"
V-56-08	Florian Birkenfeld - Kleine Klassen und gute Luft - Warum sind die Grundschulen auf dem Land besser?
V-57-08	Johann Graf Lambsdorff – The Organization of Anticorruption – Getting Incentives Right!
V-58-09	Oliver Farhauer und Alexandra Kröll - Verfahren zur Messung räumlicher Konzentration und regionaler Spezialisierung in der Regionalökonomik
V-59-09	Oliver Farhauer und Alexandra Kröll - Die Shift-Share-Analyse als Instrument der Regional- und Clusterforschung
V-60-10	Johann Graf Lambsdorff – Deterrence and Constrained Enforcement – Alternative Regimes to Deal with Bribery
V-61-10	Johann Graf Lambsdorff – Who Accepts Bribery? Evidence from a Global Household Survey
V-62-10	Oliver Farhauer und Alexandra Kröll – What We Can and What We Can't Say About Employment Growth in Specialised Cities
V-63-11	Johann Graf Lambsdorff, Manuel Schubert und Marcus Giamattei – On the Role of Heuristics - Experimental Evidence on Inflation Dynamics
V- 64-12	Manuel Schubert und Johann Graf Lambsdorff – On the Costs of Kindness - An Experimental Investigation of Guilty Minds and Negative Reciprocity
V-65-12	Manuel Schubert – Deeds rather than Omissions - How Intended Consequences Provoke Negative Reciprocity

- V-66-12 Agnes Binagwaho, Renate Hartwig, Denyse Ingeri und Andrew Makaka Mutual Health Insurance and its Contribution to Improving Child Health in Rwanda
- V-67-15 Marcus Giamattei Cold Turkey vs. Gradualism Evidence on Disinflation; Strategies from Laboratory Experiment
- V-68-15 Marcus Giamattei, Johann Graf Lambsdorff classEx an Online Software for Classroom Experiments
- V-69-15 Johann Graf Lambsdorff Preventing Corruption by Promoting Trust Insights from Behaviorial Science
- V-70-16 Lukas Schötz Power and Rents political influence on rent extraction. Quasiexperimental evidence for Bavaria
- V-71-16 Lukas Schötz Mayor Games in Bavaria: Self Selection of Local Politicians is not influenced by constitutionally defined remuneration increases
- V-72-16 Johann Graf Lambsdorff, Marcus Giamattei, Katharina Werner, Manuel Schubert-Emotion vs. Cognition – Experimental Evidence on Cooperation from the 2014 Soccer World Cup
- V-73-16 Katharina Werner Whom do People Trust after a Violent Conflict? -Experimental Evidence from Maluku, Indonesia
- V-74-16 Katharina Werner, Johann Graf Lambsdorff Emotional Numbing and Lessons Learned after a Violent Conflict – Experimental Evidence from Ambon, Indonesia
- V-75-17 Ramona Molitor Publicly Announced Speed Limit Enforcement and Its Impact on Road Safety – Evidence From the German Blitzmarathons
- V-76-17 Johannes Lorenz Population Dynamics of Tax Avoidance with Crowding Effects
- V-77-19 Katrin Huber The role of the career costs of children for the effect of public child care on fertility and maternal employment
- V-78-19 Markus Fritsch, Andrew Adrian Yu Pua, Joachim Schnurbus Revisiting habits and heterogeneity in demands
- V-79-20 Susanna Grundmann Do just deserts and competition shape patterns of cheating?
- V-80-20 Susanna Grundmann, Marcus Giamattei, Johann Graf Lambsdorff On the downward rigidity of wages evidence from an experimental labour market with monetary neutrality